Author's response to reviews

Title: Endoscopically assisted procedure for removal of a foreign body from the maxillary sinus and contemporary endodontic surgical treatment of the tooth.

Authors:

Fabio F Costa (maxil2@med.uniud.it)
Massimo M Robiony (massimo.robiony@med.uniud.it)
Corrado C Toro (maxil2@med.uniud.it)
Salvatore S Sembronio (maxil2@med.uniud.it)
Massimo M Politi (m.politi@med.uniud.it)

Version: 2 Date: 19 July 2006

Author's response to reviews:

Authors' contributions:
MP and CT designed the study. SS contributed to writing the paper. FC and MR performed surgery and wrote the main part of the paper. All authors gave useful comment on the text of the manuscript. Written consent was obtained from the patient or their relative for publication of study.

Competing Interests:
All the authors declared that no competing interests exist.